Author's response to reviews

Title: Phosphodiesterase 5 inhibitors lower both portal and pulmonary pressure in portopulmonary hypertension. A case report

Authors:

Hinrich C Bremer (hinrich-bremer@uniklinik-freiburg.de)
Wolfgang Kreisel (wolfgang.kreisel@uniklinik-freiburg.de)
Kai Roecker (kai.roecker@uniklinik-freiburg.de)
Michael Dreher (michael.dreher@uniklinik-freiburg.de)
Daniel Koenig (daniel.koenig@uniklinik-freiburg.de)
Anna-Katharina Kurz-Schmieg (kurz@medizin.ukl.uni-freiburg.de)
Hubert E Blum (hubert.blum@uniklinik-freiburg.de)
Martin Roessle (martin-roessle@t-online.de)
Peter Deibert (peter.deibert@uniklinik-freiburg.de)

Version: 6 Date: 6 July 2007

Author's response to reviews:

Thank you for your email from today.

The list of authors in the manuscript is now written exactly as they are in the submission system. There was a misspelling, sorry for that.

The corresponding author is now indicated with a superscript symbol (S)

The table is now moved to the final section of the manuscript. As we understood, an upload of the separated table is not necessary.